Author’s response to reviews

Title: The Frailty Intervention Trial (FIT)

Authors:

Nicola Fairhall (nikifairhall@bigpond.com)
Christina Aggar (c.aggar@usyd.edu.au)
Susan E Kurrle (kurrle@bigpond.com)
Catherine Sherrington (csherrington@med.usyd.edu.au)
Stephen Lord (s.lord@powmri.edu.au)
Keri Lockwood (kalockwood@nsccahs.health.nsw.gov.au)
Noeline Monaghan (n.monaghan@usyd.edu.au)
Ian D Cameron (ianc@mail.usyd.edu.au)

Version: 3 Date: 17 September 2008

Author’s response to reviews:

We have chosen to submit this research protocol to BMC Geriatrics because it is a pragmatic randomised trial of treatment of a key geriatric syndrome - frailty. The study has ethics approval and a grant from a major funding body, the Australian National Health and Medical Research Council. However, it should be noted that this research funding was for a research program from which this randomised trial has been developed.